%
% epj2003_ref.tex -- generated by sortref-2.3.3 on
% Fri Oct 31 19:19:21 CET 2003
%
%1
\bibitem{nlo}
S.~Dawson, \npb{359}{1991}{283}
%%CITATION = NUPHA,B359,283;%%
A.~Djouadi, M.~Spira, P.~Zerwas,
\plb{264}{1991}{440}.
%%CITATION = PHLTA,B264,440;%%

%\cite{Harlander:2003ai}
%2
\bibitem{Harlander:2000mg}
R.~Harlander,
\plb{492}{2000}{74}.
%%CITATION = HEP-PH 0007289;%%

%3
\bibitem{gghnnlo}
R.~Harlander, W.~Kilgore, \prl{88}{2002}{201801};
%%CITATION = HEP-PH 0201206;%%
C.~Anastasiou, K.~Melnikov,
\npb{646}{2002}{220};
%%CITATION = HEP-PH 0207004;%%
V.~Ravindran, J.~Smith, W.~van Neerven,
\npb{665}{2003}{325}.
%%CITATION = HEP-PH 0302135;%%

%4
\bibitem{gghresum}
S.~Catani, D.~de Florian, M.~Grazzini, P.~Nason,
\jhep{0307}{2003}{028};
%%CITATION = HEP-PH 0306211;%%
A.~Kulesza, G.~Sterman, W.~Vogelsang,
\arxiv{hep-ph/0309264}.

%5
\bibitem{Spira:1995rr}
M.~Spira, A.~Djouadi, D.~Graudenz, P.~Zerwas,
\npb{453}{1995}{17}.
%%CITATION = HEP-PH 9504378;%%

%6
\bibitem{higlu}
M.~Spira, \arxiv{hep-ph/9510347}.

%\cite{Harlander:2003bb}
%7
\bibitem{bill}
B.~Kilgore, talk at DPF\,2003.

%\cite{Harlander:2000mg}
%8
\bibitem{Harlander:2003bb}
R.~Harlander, M.~Steinhauser,
\plb{574}{2003}{258}.
%%CITATION = HEP-PH 0307346;%%

%\cite{Spira:1995rr}
%9
\bibitem{Dawson:1996xz}
S.~Dawson, A.~Djouadi, M.~Spira,
\prl{77}{1996}{16}.
%%CITATION = HEP-PH 9603423;%%


%10
\bibitem{gghsoft}
R.~Harlander, W.~Kilgore,
\prd{64}{2001}{013015};
%%CITATION = HEP-PH 0102241;%%
S.~Catani, D.~de Florian, M.~Grazzini,
\jhep{0105}{2001}{025}.
%%CITATION = HEP-PH 0102227;%%


%11
\bibitem{Harlander:2003kf}
R.~Harlander, M.~Steinhauser,
\arxiv{hep-ph/0308210}, {\it Phys.~Rev.}~{\bf D}, in print.
%%CITATION = HEP-PH 0308210;%%

%12
\bibitem{Harlander:2003ai} 
R.~Harlander, W.~Kilgore,
\prd{68}{2003}{013001}.
%%CITATION = HEP-PH 0304035;%%

%\cite{Boos:2003yi}
%13
\bibitem{bbhnlo}
F.~Maltoni, Z.~Sullivan, S.~Willenbrock,
\prd{67}{2003}{093005};
%%CITATION = HEP-PH 0301033;%%
E.~Boos, T.~Plehn,
\arxiv{hep-ph/0304034}.
%%CITATION = HEP-PH 0301033;%%

%\cite{Dittmaier:2003ej}
%14
\bibitem{Dittmaier:2003ej}
S.~Dittmaier, M.~Kr\"amer, M.~Spira,
\arxiv{hep-ph/0309204}.
%%CITATION = HEP-PH 0309204;%%
%15
\bibitem{pseudo}
R.~Harlander, W.~Kilgore,
\jhep{0210}{2002}{017};
%%CITATION = HEP-PH 0208096;%%
C.~Anastasiou, K.~Melnikov,
\prd{67}{2003}{037501}.
%%CITATION = HEP-PH 0208115;%%

%16
\bibitem{Brein:2003wg}
O.~Brein, A.~Djouadi, R.~Harlander,
\arxiv{hep-ph/0307206}
%%CITATION = HEP-PH 0307206;%%

